Author’s response to reviews

Title: Effectiveness of Photodynamic Therapy for Mammary and Extra-mammary Paget’s Disease: A state of the science review

Authors:

Alexa A Nardelli (alexan@ualberta.ca)
Tania Stafinski (tania@ualberta.ca)
Devidas Menon (dev.menon@ualberta.ca)

Version: 3 Date: 9 June 2011

Author’s response to reviews:

Please find attached a revised copy of the manuscript entitled “Effectiveness of Photodynamic Therapy for Mammary and Extra-mammary Paget’s Disease: A state of the science review”, authored by Alexa Nardelli, Tania Stafinski and myself. The revised manuscript addresses the formatting revisions that were requested.

Major revisions

Shading was removed from the tables in the manuscript and additional files and the text was bolded in its place.

Minor revisions

Tracking and coloured text was reformatted to normal text.
Vertical lines were removed from the tables in the manuscript and additional files.